Author’s response to reviews

Title: beta2-microglobulin induce epithelial-mesenchymal transition in human renal proximal tubule epithelial cells

Authors:

Ai qing Zhang (njaiqing@njmu.edu.cn)
Bin Wang (wangbinhewei@126.com)
Min Yang (yangmin79@sina.com)
Hui min Shi (pjr1951@163.com)
Wei hua Gan (weihuagan@njmu.edu.cn)

Version: 2
Date: 10 September 2014

Author’s response to reviews: see over
Dear Editor:

Thank you for your kind letter of “Decision on manuscript MS: 1931991443142101 BMC Nephrology” on September 9th, 2014. We revised the manuscript in accordance with your comments.

Here below is our description on revision according to your comments.

1. A separate conclusion section after the discussion has been added in the revised manuscript.

2. The names, institutions, countries and email addresses of all authors, and the full postal address of the submitting author have been added in the revised title page.

3. Page numbers and line numbering are included in the main text file of our revised manuscript.

4. Methods: Done accordingly.

5. Abbreviations used in the text have been defined in the text at first use, and a list of abbreviations was provided.

6. The authors declare that they have no competing interests.

7. An Authors' contributions section before the Acknowledgements and Reference list have been added in the revised manuscript.

8. An Acknowledgements section has been added in the revised manuscript.

Thank you for the opportunity to resubmit our revised manuscript. We look forward to hearing from you.
Sincerely yours

Bin Wang

July 19th, 2014